Author's response to reviews

Title: Tumor location and patient characteristics of colon and rectal adenocarcinomas in relation to survival and TNM classes

Authors:

Kari Hemminki (k.hemminki@dkfz.de)
Irene Santi (i.santi@dkfz.de)
Marianne Weires (m.weires@dkfz.de)
Hauke Thomsen (h.thomsen@dkfz.de)
Jan Sundquist (j.sundquist@med.lu.se)
Justo Lorenzo Bermejo (J.Lorenzo@Dkfz-Heidelberg.de)

Version: 9 Date: 20 December 2010

Author's response to reviews:

Page 1
blank after Sundquist(1,2), has to be deleted
Bermejo(1,4) instead Bermejo(1.4)
line 8: One blank before 3 Stanford has to be deleted

Page 5:
second paragraph, line 11: after (26) blank has to be deleted

Page 7, second paragraph, line 6.
The correct version is:
stage (1.45 vs 1.19, rectum is reference,1.00).

Page 9:
second paragraph, line 10: Thus, - a comma has to be set

Table 1: the second line of the title should not be centered but with an indent
Table 3: as in Table 1